Author's response to reviews

Title: Clinical Utility of Antinuclear Antibody Tests in Children

Authors:

Julie L. McGhee (julie-mcghee@ouhsc.edu)
Lauren M. Kickingbird (lauren-kickingbird@ouhsc.edu)
James N. Jarvis (james-jarvis@ouhsc.edu)

Version: 2 Date: 7 July 2004

Author's response to reviews:

We appreciate the opportunity to resubmit this paper. We have added the data on complete blood counts in children with SLE as suggested by the reviewer.